Author's response to reviews

Title: Evaluation of the Middle Cerebral Artery Occlusion Techniques in the Rat by in-vitro 3-Dimensional Micro- and Nano Computed Tomography

Authors:

Alexander C Langheinrich (alexander.langheinrich@radiol.med.uni-giessen.de)
Mesut Yeniguen (Mesut.Yeniguen@neuro.med.uni-giessen.de)
Anne Ostendorf (Anne.M.Ostendorf@med.uni-giessen.de)
Simone Doenges (S.Marhoffer@gmx.de)
Marian Kampschulte (Marian.Kampschulte@radiol.med.uni-giessen.de)
Georg Bachmann (g.bachmann@kerckhoff-klinik.de)
Erwin Stolz (Erwin.Stolz@neuro.med.uni-giessen.de)
Tibo Gerriets (tibo.gerriets@neuro.med.uni-giessen.de)

Version: 3 Date: 13 April 2010

Author's response to reviews: see over
Dear Editor,

This is a re-submission of our revised manuscript entitled

“Evaluation of the Middle Cerebral Artery Occlusion Techniques in the Rat by in-vitro 3-Dimensional Micro- and Nano Computed Tomography”.

The authors thank you and the reviewers for the comments and recommendations. We answered concerns on a point-by-point basis and have indicated all changes made in the manuscript.

We hope that the present version of the manuscript is now acceptable for publication in your journal.

Yours sincerely,

Alexander C. Langheinrich
Comments to the Editor:

The section „Competing interests“ and „Authors‘ contributions“ have been implemented into the revised version of the manuscript. We highlighted the sections in „red“. 